Author's response to reviews

Title: Growth in Children with Cerebral Palsy during five years after Selective Dorsal Rhizotomy: a practice-based study

Authors:

Lena Westbom (lena.westbom@med.lu.se)
Annika Lundkvist Josenby (annika.lundkvist@med.lu.se)
Philippe Wagner (pw@nko.se)
Eva Nordmark (eva.nordmark@med.lu.se)

Version: 3 Date: 28 June 2010

Author's response to reviews:

Dear editorial production team,

I have tried several times to upload our manuscript updated according to your letter June 23rd, but have not yet succeeded. My computers says there is something wrong with the page. I will try again in the end of this session and if nessesary try again tomorrow morning.

The nine figures were uploaded in power-point format and I cannot change them. We have used the same scales from +6 to -6 in all figures to make them easily comparable. If they have to be cropped outside the scales I hope you can do it, as we are unable to open the format you sent them in.

The tables are now included in the manus, after references and before figure legends. I have made all the other formatting changes according to your "Major revisions". Author names, square brackets to all references done, the order of references corrected

Best regards,

Lena Westbom